Author’s response to reviews

Title: Revitalizing physical assessment in undergraduate nursing education - what skills are important to learn, and how are these skills applied during clinical rotation? A cohort study

Authors:

H. Ösp Egilsdottir (osp.egilsdottir@usn.no)

Kirsten Røland Byermoen (kirsten.roland.byermoen@usn.no)

Anne Moen (anne.moen@medisin.uio.no)

Hilde Eide (hilde.eide@usn.no)

Version: 3 Date: 22 Jul 2019

Author’s response to reviews:

Dear Editor in BMC Nursing,

We will thank you for giving us the opportunity to revise and resubmit our manuscript entitled “Revitalizing physical assessment in undergraduate nursing education - What skills are important to learn, and how are these skills applied during clinical rotation? A cohort study” in BMC Nursing.

We have clarified better what was needed regarding declarations in the article and are point by point elaborated in the attached document.